Author’s response to reviews

Title: EFFECTIVENESS OF PHYSICAL EXERCISE IN THE TREATMENT OF DEPRESSION IN OLDER ADULTS AS AN ALTERNATIVE TO ANTIDEPRESSANT DRUGS IN PRIMARY CARE

Authors:

Jesús López-Torres Hidalgo (jesusl@sescam.org)
Luis Aguilar Salmerón (geniusluis10@gmail.com)
Clotilde Boix Gras (cboix@sescam.org)
Monchi Campos Rosa (monchi.campos@gmail.com)
Francisco Escobar Rabadán (fescobarr@yahoo.es)
Concepción Escolano Vizcaíno (concepcione@sescam.jccm.es)
José Luis Estellés Belenguer (jlestelles@sescam.jccm.es)
Juan Fernández Martín (jfernandezm@jccm.es)
Vicente Ferrer López (ferrerlopezv@gmail.com)
Almudena Legido García (alegido@sescam.jccm.es)
Jaime López-Torres López (jaime.lopeztorres.lopez@gmail.com)
Jesús López-Torres López (jlopeztl@yahoo.es)
María Ángeles López Verdejo (malopezv@sescam.jccm.es)
Ana López Yeste (anloye@sescam.jccm.es)
María Ángeles Lloret Callejo (malloret@sescam.org)
María Jesús Montes Lozano (mjmlxi@gmail.com)
Juana Muñoz Núñez (jmunozn@sescam.jccm.es)
Karen Nieto Rodríguez (Knnieto@yahoo.com)
Joseba Rabanales Sotos (Joseba.Rabanales@uclm.es)
RESPONSE LETTER

1. The individual contributions of ALL authors to the manuscript should be specified in the Authors’ Contributions section. Guidance and criteria for authorship can be found here: http://www.biomedcentral.com/submissions/editorial-policies#authorship.

Please represent authors' names using their full initials, not their full/sur- name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors. There are currently several duplicates.

In the new version of the manuscript all authors are represented by their initials and there are no duplicate authors.

2. Please add all authors to the submission system.

All the authors have been added to the submission system.

3. At this stage, please upload your manuscript as a single, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

The new version of the manuscript has been a single and clean version.